Author's response to reviews

Title: Gender differences in the utilisation of health-care services among the older adult population of Spain

Authors:

Aurea Redondo Sendino (aurea.redondo@uam.es)
Pilar Guallar Castillon (mpilar.guallar@uam.es)
Jose Ramon Banegas Banegas (joseramon.banegas@uam.es)
Fernando Rodriguez Artalejo (fernando.artalejo@uam.es)

Version: 7 Date: 15 June 2006

Author's response to reviews:

We send a revised version of the manuscript where we have made the following changes suggested by the Editorial production team in their mail of 15 June, 2006:

a) Section headings - We have used sentence case for all headings and sub-headings in the manuscript (we have removed all unnecessary capitals). For example Abstract not ABSTRACT.

b) References - In the manuscript text, we have placed the reference before the full stop.